Author's response to reviews

Title: Antimicrobial resistance predicts death in Tanzanian children with bloodstream infections: a prospective cohort study

Authors:

Bjorn Blomberg (bjorn.blomberg@med.uib.no)
Karim P Manji (kmanji@muchs.ac.tz)
Willy K Urassa (wurassa@muchs.ac.tz)
Bushir S Tamim (tambs2000@yahoo.com)
Davis SM Mwakagile (dmwakagile@muchs.ac.tz)
Roland Jureen (Roland_Jureen@Alexhosp.com.sg)
Viola Msangi (vmsangi@muchs.ac.tz)
Marit G Tellevik (marit.gjerde@med.uib.no)
Mona Holberg-Petersen (mona.holberg-petersen@ulleva.no)
Stig Harthug (stig.harthug@helse-bergen.no)
Samwel Y Maselle (smaselle@muchs.ac.tz)
Nina Langeland (nina.langeland@helse-bergen.no)

Version: 5 Date: 20 May 2007

Author's response to reviews:

Dear Editor,

We are grateful that BMC Infectious Diseases has accepted our manuscript "Antimicrobial resistance predicts death in Tanzanian children with septicemia" for publication. Please find enclosed a revised version of our article, which has been updated to agree with the suggested formatting changes.

1. Text in the "Header" and "Footer" fields has been removed.
2. Keywords and word counts have been removed.
3. The wording of the "Competing interest" section has been amended
4. Figure titles have been removed from the figure files, and added in the manuscript file, as requested. There are no figure legends in this version.
5. The figures have been renumbered as Figure 1 through 6 (in stead of 1a through 1f), as requested, and the references to these figures in the manuscript have been updated.
6. The tables have been reformatted as suggested.

Furthermore, we have proofread the manuscript.

Please advise us if further changes are required.

Yours sincerely,

Bjorn Blomberg